Author's response to reviews

Title: Estimating the workload associated with symptoms-based ovarian cancer screening in primary care: an audit of electronic medical records

Authors:

Anita WW Lim (a.lim@qmul.ac.uk)
David Mesher (david.mesher@phe.gov.uk)
Peter Sasieni (p.sasieni@qmul.ac.uk)

Version: 2 Date: 30 July 2014

Author's response to reviews:

Inserted line numbers.
Removed page breaks (replaced with section breaks)
Moved ethics statement into methods.
Renamed "Introduction" as "Background"
Updated references: Removed one reference and re-formatted online references to include URL and date accessed.